Author's response to reviews

Title: The influence of P-glycoprotein expression and its inhibitors on the distribution of doxorubicin in breast tumors

Authors:

Krupa J Patel (krupa.patel@utoronto.ca)
Ian F Tannock (Ian.Tannock@uhn.on.ca)

Version: 6 Date: 30 September 2009

Author's response to reviews:

Dear Editors,

Thank you once again for your decision to accept our manuscript for publication. Final slight revisions were made to correct grammatical mistakes on page 4 (top paragraph), page 8 and 12 (changed ‘ten’ to ‘10’), page 12 (second paragraph), page 13 (first paragraph), page 15 (first paragraph), and page 17 (Authors contributions) and in Figure legends (figure 2 and figure 3).

We appreciate you guidance and assistance and if there is any other information you require please let us know,

Krupa J. Patel
Ian F. Tannock